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CERN status

• HV cage has been re-installed.
• HV module controller will be tested 

today.
• Elena has problem until last week. 

(H- beam and e- cooler)

• Linac problem has been solved.
• Cable will be prepared before August.

• Proton beam has been tested in the 
outside of GBAR zone.


